

I, along with my co-authors, wish to re-submit the manuscript titled “Pancreatitis, panniculitis, and polyarthritis syndrome caused by pancreatic pseudocyst: A case report”.

We thank the reviewers for thoughtful suggestions and insights. The manuscript has benefited from these insightful suggestions.

The manuscript has been rechecked and the necessary changes have been made in accordance with the reviewers’ suggestions and have been indicated in highlighted text. The responses to all comments have been prepared and attached herewith.

Reviewer 1

If the exact cause is found, we can discuss it, but it’s still unknown. There is not much research on the exact cause of this syndrome, there are only a few older ones, and they do not guess the exact cause. It is likely that further research will be needed to find the cause.

Reviewer 2

The difference with this case report is that it is the first report of PPP syndrome, forming a fistula with IVC. In addition, we would like to report the experience of unexpected bleeding during surgery because we could not identify the fistula before surgery, and hope to prevent any complications during the surgery.

Also, the revision was made in consideration of the points indicated by the reviewer.